Author’s response to reviews

Title: Suppression of the Epithelial-Mesenchymal Transition by SHARP1 Is Linked to the NOTCH1 Signaling Pathway in Metastasis of Endometrial Cancer

Authors:

Yun Liao (liao.yun@aliyun.com)
Xiaoying He (hexiaoying06@gmail.com)
Haifeng Qiu (haifengqiu120@163.com)
Qi Che (cheqi1477@126.com)
Fangyuan Wang (sunflowerwang@hotmail.com)
Wen Lu (wenwenyou2002@126.com)
Zheng Chen (chenzheng_cynthia@hotmail.com)
Meiting Qiu (xiaomei727@hotmail.com)
Jingyun Wang (fishann0303@gmail.com)
Huihu Wang (clara.wang@hotmail.com)
Xiaoping Wan (wanxp@sjtu.edu.cn)

Version: 3
Date: 29 June 2014

Author’s response to reviews:

June 29, 2014

Dear editors and Annie Lyn Bravo:

Thank you very much for accepting our manuscript. After minor revision, we would like to resubmit the revised manuscript entitled “Suppression of the Epithelial-Mesenchymal Transition by SHARP1 Is Linked to the NOTCH1 Signaling Pathway in Metastasis of Endometrial Cancer” to BMC Cancer (MS: 1897847224122405).

The permit number of our animal experiments, as specified by reviewer 2, has been addressed in page 6, paragraph 1. We would like to express our great appreciation to you and to the reviewers for the helpful comments on our paper.

Thank you for accepting this manuscript. We are looking forward to hearing from you.

Best regards,

Yun Liao

Corresponding author:

Dr. Xiaoping Wan

Department of Obstetrics and Gynecology, Shanghai First Maternity and Infant Hospital Affiliated to Tong Ji University, Shanghai, China.

Fax: +86 21 54032482; E-mail: wanxp@sjtu.edu.cn